Author's response to reviews

Title: Home: The place the older adult can not imagine living without

Authors:

Catharina Gillsjo (catharina.gillsjo@his.se)
Donna Schwartz-Barcott (dsb@uri.edu)
Irene von Post (irene.von.post@telia.com)

Version: 10 Date: 9 March 2011

Author's response to reviews:

Dear editorial production team,

Their was found some bolded periods which were changed to normal format.

Best regards,

Catharina Gillsjö